Author’s response to reviews

Title: The effectiveness and cost effectiveness of the PAtient-Centred Team (PACT) model: study protocol of a prospective matched control before-and-after study

Authors:

Trine Bergmo (trine.bergmo@telemed.no)
Gro Berntsen (gro.bernsen@telemed.no)
Monika Dalbakk (Monika.Marie.Dalbakk@unn.no)
Markus Rumpsfeld (markus.rumpsfeld@unn.no)

Version: 2 Date: 07 Oct 2015

Author’s response to reviews:

Response to the comments

- title should be changed to "The effectiveness and cost effectiveness of the PAtient-Centred Team (PACT) model: study protocol of a prospective matched control before-and-after study"

Response:

I have changed the title as suggested.

- the emails of all authors should be added in the title page of the manuscript

Response:

I have included the emails of all authors on the front page:
trine.bergmo@telemed.no
gro.berntsen@telemed.no
monika.marie.dalbakk@unn.no
markus.rumpsfeld@unn.no

I have highlighted the changes in the document.

I have followed the instructions for authors and adhered to guidelines for protocol papers. I have included authors’ contribution at the end of the manuscript.
The manuscript has been proof read by a professional language editing service Scribendi.